Author's response to reviews

Title: Characterization of medical students recall of factual knowledge using learning objects and repeated testing in a novel e-learning system

Authors:

Tiago Taveira-Gomes (tiago.taveira@me.com)
Rui Prado-Costa (rcosta4540@gmail.com)
Milton Severo (milton@med.up.pt)
Maria Amélia Ferreira (mameliaferreira@med.up.pt)

Version: 2 Date: 16 July 2014

Dear Editorial Board,

Regarding the changes requested on my manuscript prior to the peer review, I submit my manuscript with the following changes:
- added line numbers
- added the email address of all authors

Best regards,

Tiago Taveira Gomes